Author’s response to reviews

Title: How can we get more people with long-term health conditions involved in parkrun? A qualitative study evaluating parkrun's PROVE project

Authors:

Helen Quirk (h.quirk@shu.ac.uk)
Steve Haake (s.j.haake@shu.ac.uk)

Version: 2 Date: 27 Aug 2019

Author’s response to reviews:

1. I can confirm that all people listed in the acknowledgements section of the Declarations have agreed to be named.

2. I have uploaded the clean version of the manuscript (no tracked changes, comments, highlights, strikethroughs or text in different colours) named revision 2.